Author's response to reviews

Title: Comparison of pulmonary and extrapulmonary tuberculosis in Nepal- a hospital-based retrospective study

Authors:

Chandrashekhar T Sreeramareddy (chandrashekharats@yahoo.com)
Kishore V Panduru (pandu_vki@yahoo.com)
Sharat C Verma (verma_sharat@yahoo.com)
Hari S Joshi (drijoshharish@rediffmail.com)
Michael N Bates (m_bates@berkeley.edu)

Version: 5 Date: 17 January 2008

Author's response to reviews:

Dear Editorial Team,

The following changes were made to this revised manuscript.
1) The author affiliation details were revised.
2) The Sources of funding was removed/deleted.
3) The reference number 26 was updated i.e. all the authors were listed.
4) Figures were removed from the main manuscript except the figure legends.